Author's response to reviews

Title: Prognosis of West Nile Virus Associated Acute Flaccid Paralysis: A Case Series

Authors:

Jennie Johnstone (johnsj48@mcmaster.ca)
Steven Hanna (hannas@mcmaster.ca)
Lindsay Nicolle (LNicolle@exchange.hsc.mb.ca)
Michael Drebot (Mike.Drebot@phac-aspc.gc.ca)
Binod Neupane (binodn@gmail.com)
James Mahony (mahonyj@mcmaster.ca)
Mark Loeb (loebm@mcmaster.ca)

Version: 6 Date: 15 June 2011

Author's response to reviews:

Dear JMCR Editorial Team,

Thank you for giving us the opportunity to revise and resubmit our manuscript entitled, “Prognosis of West Nile Virus Associated Acute Flaccid Paralysis: A Case Series” (MS: 4429529750551201 R3). We have made significant formatting changes. We have restructured our case series so that each of the 7 cases are now discussed separately. We have included a 'summary of cases' section as well; however this could be omitted if it is not in keeping with the journal's style.

Respectfully submitted on behalf of the authors,

Jennie Johnstone

Jennie Johnstone, MD
Assistant Professor,
Division of Infectious Diseases,
Department of Medicine,
McMaster University